Week 23 starting 5June 06
SM18 Operation Meeting (Tuesdays) 6 June 06

Bench status  on Monday  05-06-2006 at:  Midday
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	  TBA1
	MBBR1347
	ICS 4 Final connection
	300 K


	TBA2
	MBAR2314
	PT 14 Magnet MAPS
	300 K


	TBB1
	SSS038
	PT 5.2 1.5 KA Quench HF
	1.9 K


	TBB2
	SSS626
	FINISHED SINCE SAT 3rd June morning 
	300 K 2P2M


	TBC1
	MBAL2341
	PT 14 Magnet MAPS
	300 K


	TBC2
	MBAR2330
	Prep 5 Cool Down
	Cooling


	TBD1
	SSS633
	S4 Field Strength
	4.4 K


	TBD2
	SSS690
	FINISHED since Friday 2nd June 2300 hrs 
	300 K
2P2M


	TBE1
	MBAL2340
	PT 13.2 Resist. Meas.
	300 K


	TBE2
	MBAR2273
	Prep 5 Cool Down
	Cooling

	TBE2
	MBAR2273
	Prep 5 Cool Down
	Cooling


	TBF1
	MBBR1351
	Prep 5 Cool Down
	Cooling


	TBF2
	MBBR2262
	PT 6.2 Training 2
	1.9 K
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  CONFIG : 
 D8- AS2 –IRSSS2;
Supervisor: nick/        Eq support : 16-3602 :

18 Magnet Tests completed :  11 Dipoles (1  MM),    2 Arc-SSS  ( no   REPEAT)

(16 Magnets in all)

1  500-series ( 505 (  

                                                  2  600-series [ 647 (313 hrs),  677 (154 hrs) ]

2284 repeated for cryo reason only; 1353 ROBTC                              
General Points :

· Last weekend problems  (  Magnets Not removed on D2, B2
· For whole weekend !!

· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· News /Situation from (SSS, SSSS)  :  J Halik
· Situation SSSS  :OP issues : 
· A.O.B.

Role of Shift Leaders & Visits

Role of Shift Leaders & Special Experiments

Role of Shift Leaders &  ICS, Cryo in SMCR etc

Bench status  on Monday  05-06-2006 at:  Midday
	Magnets under Test on 05-06 11:52:31 AM 

	-
	Bench
	Magnet
	Sh.
	Tested
	Quench
	Start
	Connect
<24h
	Cooling
<26h
	Cold Test
<36h
	Warming
<12h
	Disconnect
<12h
	Total
	
	
	

	1
	TBA1
	MBBR1347
	[image: image4.png]



	1 
	0 
	03-06
	43h[image: image5.png]



	-
	-
	-
	-
	43 h.
	
	
	

	2
	TBA2
	MBAR2314
	[image: image6.png]



	1 
	1 
	31-05
	2750h
	Check Valueh
	2803h
	17h
	10h
	2846 h.
	
	
	

	3
	TBB1
	SSS038
	[image: image7.png]



	4 
	0 
	01-06
	51h
	20h
	13h[image: image8.png]



	-
	-
	84 h.
	
	
	

	4
	TBB2
	SSS626
	[image: image9.png]



	1 
	1 
	29-05
	31h
	35h
	25h
	7h
	51h
	164 h.
	
	
	

	5
	TBC1
	MBAL2341
	[image: image10.png]



	1 
	3 
	31-05
	22h
	29h
	24h
	15h
	11h
	109 h.
	
	
	

	6
	TBC2
	MBAR2330
	[image: image11.png]



	1 
	0 
	04-06
	45h
	4h[image: image12.png]



	-
	-
	-
	49 h.
	
	
	

	7
	TBD1
	SSS633
	[image: image13.png]



	1 
	2 
	01-06
	47h
	12h
	32h[image: image14.png]



	-
	-
	91 h.
	
	
	

	8
	TBD2
	SSS690
	[image: image15.png]



	1 
	1 
	29-05
	27h
	0h
	45h
	14h
	66h
	167 h.
	
	
	

	9
	TBE1
	MBAL2340
	[image: image16.png]



	1 
	3 
	01-06
	24h
	23h
	18h
	15h
	0h
	84 h.
	
	
	

	10
	TBE2
	MBAR2273
	[image: image17.png]



	1 
	0 
	02-06
	38h
	26h[image: image18.png]



	-
	-
	-
	64 h.
	
	
	

	11
	TBF1
	MBBR1351
	[image: image19.png]



	1 
	0 
	03-06
	23h
	20h[image: image20.png]



	-
	-
	-
	43 h.
	
	
	

	12
	TBF2
	MBBR2262
	[image: image21.png]



	1 
	2 
	02-06
	37h
	28h
	15h[image: image22.png]



	-
	-
	80 h.
	
	
	


.
	Magnets under Test from 28-05-2006 to 03-06-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBL1236 
	[image: image23.png]



	


	23-05-2006 
	48 h.
	27 h.
	128 h.
	35 h.
	21 h.
	03-06-2006 
	259 h.
	Stripping
	
	

	2
	TBA2
	MBAR1354 
	[image: image25.png]



	


	24-05-2006 
	34 h.
	58 h.
	25 h.
	20 h.
	12 h.
	31-05-2006 
	149 h.
	Stripping
	
	

	3
	TBB1
	SSS086 
	[image: image27.png]



	


	29-05-2006 
	31 h.
	11 h.
	19 h.
	9 h.
	9 h.
	01-06-2006 
	79 h.
	Stripping
	
	

	4
	TBB1
	SSS264 
	[image: image29.png]



	


	25-05-2006 
	33 h.
	15 h.
	29 h.
	9 h.
	10 h.
	29-05-2006 
	96 h.
	Stripping
	
	

	5
	TBB2
	SSS505 
	[image: image31.png]



	


	22-05-2006 
	36 h.
	13 h.
	36 h.
	10 h.
	85 h.
	28-05-2006 
	180 h.
	Stand By
	
	

	6
	TBC1
	MBAR2310 
	[image: image33.png]



	


	27-05-2006 
	24 h.
	23 h.
	22 h.
	15 h.
	11 h.
	31-05-2006 
	95 h.
	Stripping
	
	

	7
	TBC2
	MBAR1318 
	[image: image35.png]



	


	30-05-2006 
	20 h.
	22 h.
	24 h.
	14 h.
	14 h.
	03-06-2006 
	94 h.
	Stripping
	
	

	8
	TBC2
	MBAR2272 
	[image: image37.png]



	


	26-05-2006 
	22 h.
	24 h.
	29 h.
	16 h.
	7 h.
	30-05-2006 
	98 h.
	Stripping
	
	

	9
	TBD1
	SSS647 
	[image: image39.png]



	


	19-05-2006 
	103 h.
	15 h.
	171 h.
	15 h.
	9 h.
	01-06-2006 
	313 h.
	Stand By
	
	

	10
	TBD2
	SSS677 
	[image: image41.png]



	


	22-05-2006 
	32 h.
	16 h.
	78 h.
	9 h.
	19 h.
	29-05-2006 
	154 h.
	Stand By
	
	

	11
	TBE1
	MBAR2276 
	[image: image43.png]



	


	28-05-2006 
	18 h.
	24 h.
	21 h.
	14 h.
	18 h.
	01-06-2006 
	95 h.
	Stripping
	
	

	12
	TBE1
	MBBR2284 
	[image: image45.png]



	


	23-05-2006 
	28 h.
	25 h.
	20 h.
	24 h.
	17 h.
	28-05-2006 
	114 h.
	Stripping
	
	

	13
	TBE2
	MBAR1353 
	[image: image47.png]



	


	25-05-2006 
	22 h.
	40 h.
	41 h.
	-
	-
	30-05-2006 
	103 h.
	Not Found or Thermal cycle
	
	

	14
	TBE2
	MBAR1353 
	[image: image49.png]



	


	30-05-2006 
	0 h.
	32 h.
	21 h.
	20 h.
	14 h.
	02-06-2006 
	87 h.
	Stripping
	
	

	15
	TBF1
	MBBR2279 
	[image: image51.png]



	


	22-05-2006 
	51 h.
	32 h.
	45 h.
	15 h.
	14 h.
	29-05-2006 
	157 h.
	Stripping
	
	

	16
	TBF1
	MBBR2284 
	[image: image53.png]



	


	29-05-2006 
	20 h.
	35 h.
	22 h.
	18 h.
	25 h.
	03-06-2006 
	120 h.
	Stripping
	
	

	17
	TBF2
	MBAR2327 
	[image: image55.png]



	


	29-05-2006 
	19 h.
	34 h.
	13 h.
	20 h.
	8 h.
	01-06-2006 
	94 h.
	Stripping
	
	

	18
	TBF2
	MBBR2331 
	[image: image57.png]



	


	24-05-2006 
	27 h.
	28 h.
	18 h.
	15 h.
	9 h.
	28-05-2006 
	97 h.
	Stripping
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL1236
	11821.3 A.
(D1-L-S01)
	12360 A.
(D1-L-S01)
	12784.8 A.
(D2_U-L)
	[image: image59.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR1354
	11331.9 A.
(D2_L)
	12448.9 A.
(D1_U)
	12649.8 A.
(D2_L)
	[image: image60.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS086
	11987.1 A.
(QI_12)
	12265.8 A.
(QE_12)
	12677.3 A.
(QI_12)
	[image: image61.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS264
	11312 A.
(QI_12)
	12475 A.
(QE-12)
	12771.2 A.
(QE_12)
	[image: image62.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS505
	11527.4 A.
(QE_12)
	11596.2 A.
(QI-12)
	12360.5 A.
(QE_34)
	12853 A.
[image: image63.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAR2310
	8814.2 A.
(D2_L)
	11286.2 A.
(D1_L)
	12409.3 A.
(D2_L)
	12660.7 A.
(D1_U)
	[image: image64.png]



	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAR1318
	12394.8 A.
(D1-U)
	12583.9 A.
(D1-U)
	12791.6 A.
(D1_U)
	[image: image65.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAR2272
	11009.7 A.
(D1_L)
	11683.1 A.
(D1_U)
	12197.6 A.
(D2_L)
	12475.4 A.
(D2_U)
	12789.5 A.
(D2_L)
	12851.3 A.
(D2_U)
	12851 A.
[image: image66.jpg]



	-
	-
	-
	-
	-
	-

	9
	SSS647
	3622.7 A.
(MQY_I_34)
	3759.6 A.
(I_Mqy12)
	2935 A.
(MQY-E 12)
	[image: image67.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS677
	5424.5 A.
(Int_MQMC)
	5727.9 A.
(INT_MQMC)
	5779.3 A.
(Ext_MQM)
	5820 A.
[image: image68.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR2276
	10928.7 A.
(D2_L)
	12189.2 A.
(D1_L)
	12526.6 A.
(D2_U)
	[image: image69.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBR2284
	11790 A.
(D2_U)
	12264.9 A.
(D1_U)
	12614.1 A.
(D1_U)
	[image: image70.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR1353
	11226.4 A.
(D2_U)
	11548.9 A.
(D2_U)
	11742.7 A.
(D1_L)
	11963.9 A.
(D2_L)
	12300.4 A.
(D2_U)
	12429.9 A.
(D1_L)
	12594 A.
(D2_U)
	12651.2 A.
(D1_L)
	12794.6 A.
(D1-U)
	[image: image71.png]



	-
	-
	-

	14
	MBAR1353
	12134.9 A.
(D2_L)
	12337.2 A.
(D1_L)
	12484.4 A.
(D2_U)
	[image: image72.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR2279
	9843.8 A.
(D2_U)
	12030.3 A.
(D1_L)
	12547 A.
(D2_U)
	[image: image73.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBBR2284
	12646.2 A.
(D1_U)
	[image: image74.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	17
	MBAR2327
	12627.2 A.
(D1_L)
	12850 A.
[image: image75.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	18
	MBBR2331
	11458.3 A.
(D1_L)
	12644.4 A.
(D2_L)
	12850 A.
[image: image76.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
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          Effective 16 magnets Tested this week
            2284 Repeated for Cryo, 1353 ROBTC, 1236 Long MM (marco, juan), ;  505 could Not be removed by ICS (logistics Org.)

          647 usual reasons of connections & powering +SSW,    677 had  special tests at cold by someone for beam loss? .. etc)
	Date
	Bench
	Problem
	Type
	App.
Soft.
	Action/advise

	Tuesday_30_A
	TBB1
	SSS08 : HV test front panel not opening in the control room.
	AB/CO
	HV
	To be investigated by AB/CO expert.

	Tuesday_30_A
	TBE1
	MB2276 : No cryo OK for PT4. 
	ACR
	
	ACR team contacted. They reported problem with helium level. Expected to be ok in 45 minutes.

	Tuesday_30_A
	TBA1
	MB1236: Special test on magnetic axis. The team carrying out magnetic measurements on A1 is placing the Laser in the middle of the way to D1 and doing the tests without a proper barricade. Whoever passes by they shout saying that the laser is there and we are not supposed to cross. Also they were blocking operators from carrying out tests. When they were asked by the shift leader to put barricades, the response was very casual asking us to contact Marco. 
	
	
	Marco was contacted and requested to instruct them to follow safety rules and know the rules of working in sm18.

Problem solved.

	Tuesday_30_A
	TBC2
	MB1318: Problems with the MOBIC. It does not read the bar code. Occasionally when it reads, it does not display the number in the corresponding field and the light in the scanner goes off. Tried with both the units but no success.
	MTM
	
	Finally the magnet details were entered manually.



	Tuesday_30_A
	TBD1
	SSS647: Cryo interlock appearing on TEMA while launching MM2 - LHC Cycle. Tried to Launch from the Sun station down stairs. MM2 could be launched but the magnet quenched at 808A.
	MTM
	
	Guy asked us to perform one more cleansing cycle and then try again after consulting TC. TC instructed cryo to increase He flow.
He asked us to perform training run after getting CRYO OK and if the magnet does not quench below 3000 asked the op team to try running LHC cycle again.

	Tuesday_30_A
	TBD1
	SSS647: Training launched after obtaining CRYO OK but the magnet quenched at 888A. ACR team insisted to perform one more training after implementing more corrective measures.
	MTM
	
	

	Tuesday_30_N
	TBD1
	SSS647: Problem of maintaining cryo stability at 4.5K.
	ACR
	
	

	Tuesday_30_N
	TBD1
	SSS647: Problem with magnet quenching at low current (about 800A or less) in V_bus_ext and V_bus_int. 
	MTM
	
	The Cryo informed that the problem is not coming from cryogeny.

 TC discussed with Stephane and decided to cancel the machine cycle.

Magnet warm_up. Reason is a long flexible in the N-line that makes a loop and is a reservoir of gaz bubles. Action will be taken among the MTM mechanics to watch out during the connection. 

	Wednesday_31_N
	TBA2
	MB2314 : Problem with HV rack MY92 : PLC-HV fault does not allow the test to be carried out. 


	MTM
	
	MY 93 used instead of MY92. Guy was informed; he will look into the problem tomorrow..

	Wednesday_31_N
	TBE2
	MB1353: Problem with the charge of one quench heater. 
	IT?
	
	Problem will be reported to equipment support in the morning shift. Problem solved by reset of PLCSE. Quench heater are charging now. Problem with the network?

	Thursday_01_M
	TBC1
	MB 2341 : Prep2.1 could not be launched.
	IT?
	
	Problem solved by resetting the PLCSE. Problem with the network?

	Thursday_01_A
	TBD1
	SSS633: Only magnet name label present. Anti-cryostat and other labels absent.
	CRI
	
	

	Thursday_01_N
	TBE1
	MB2340 : Mobic is not accepting the data during the scan Data for E1 and B1 entered manually in MTF.
	MTM
	
	

	Thursday_01_N
	TBB2 
	SSS626 : NC generated in CDBP for temp sensor not OK
	CRI
	
	HV fault with Tsensor.

	Thursday_01_N
	TBB1
	SSS038: ICS has informed that, the initial connection of the magnet before Lyra, could not be done completely due to insufficient lengths of some internal wires. 
	ICS
	
	Done Friday Morning by ICS.

	Thursday_01_N
	TBD2
	SSS690 : Black (-) DC cable lead (2A cable) at top of CFB is broken.
	MTM
	
	

	Thursday_01_N
	TBF2
	MB2262 : Mobic not working and anticryostat labels absent. 
	MTM/ICS
	
	Only magnet name label read and entered in MTF manually.



	Friday_02_M
	TBB2
	SSS626 : During PT10.1 warm-up quench analysis, AQA alarm appeared, so the magnet is kept on stand-by temp for consultation with TC in morning. 
	MTM
	
	TC checked the data and asked to warm up the magnet.

	Friday_2_N
	TBC1

TBA2

TBB2
	MB2341 :Interlock (No PC Permit)do not allow to lunch PT4 (slow power abort.
Same problem.

HV panel not displayed.
	AB/CO
	
	PLCSE reset but no success. Guy called.

Communication problem between PLCSE and PLCCU. After Switch off all PLCSE, Reset of PLCCU & restart. 
OK for TBA & TBB but still a specific communication problem on TBC.
RS communication error with PC 14KA. Guy tried to change all rs convertors anf rs card on PC. Finally, AB/CO said to switch to gateway communication to control the PC. 

OK then. RS problem need to be investigated on Tuesday by AB-CO and/or AB-PO.



	Saturday_03_M
	TBD1
	SSS633 :Vacuum leak detected by ACR in the anti-cryostat. 
 

	MTM
	
	TC called and expert ACR? Solved by MTM mechanics.

	Saturday_03_A
	TBD2
	SSS690: One pin of DC cable is found broken inside the socket.
	MTM
	
	Guy contacted. He made a temporary repair of the 20A DC cable connector ((-) lead). The 20A DC leads must be temporarily connected directly to the power leads. He asked to the next shifts:
· during disconnection, one should not pull the cable, but the connector. 

·  to be very careful to the fact that when disconnecting the lead, current must be off (no protection anymore)

	Saturday_03_A
	TBA2
	MB2314 : Mobic problem : In spite of having full charge and full connectivity, the scanned data could not be validated. 
	AB/CO?
	
	Comment by OP team: Frequent problems with mobic, more of a trouble than a facility, leading to manual updating of magnet and accessories info in to the MTF of late 

	Saturday_03_N
	TBA1
	MB1347: SMS from ICS to perform Lyra test. OP team observed that WP04 was not uploaded.
	ICS
	
	WP=4 (HV test) was not done? HV test failed and performed again by ICS

	Sunday_04_M
	TBD1
	SSS633 : the powering of the K-L circuit failed.
	MTM
	LF
	Guy advised to proceed with MQY training.

	Sunday_04_M
	TBC2
	Sun MTA 58 monitor screen is not working. The sun is hung.
	AB/CO
	
	Guy called and  asked OP team to re-boot the sun.

	Sunday_04_N
	TBB1
	SSS038: PT4 not working. The ramp in the power supply cannot start. PLCSE correct. Problem with FB Loop interlock.
Corrector test is not working
	AB/PO
	
	We continue with correctors and will do PT4 later 
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